Author’s response to reviews

Title: MiR-182-5p and its target HOXA9 in non-small cell lung cancer: a clinical and in-silico exploration with the combination of RT-qPCR, miRNA-seq and miRNA-chip

Authors:

Li Gao (2339352006@qq.com)
Shi-bai Yan (522937211@qq.com)
Jie Yang (yanglz2005@126.com)
Jin-liang Kong (kjl071@126.com)
Ke Shi (529057611@qq.com)
Fu-chao Ma (mafuchao@live.cn)
Lin-zhen Huang (944354533@qq.com)
Jie Luo (1164524563@qq.com)
Shu-ya Yin (349230442@qq.com)
Rong-quan He (herongquan@gxmu.edu.cn)
Xiao-hua Hu (huxiaohua@gxmu.edu.cn)
Gang Chen (chengang@gxmu.edu.cn)

Version: 3 Date: 11 Dec 2019

Author’s response to reviews:

1. Availability of data and materials

Although you mentioned that you downloaded genomes from TCGA, GEO, miRWalk and TargetScanHuman in the Availability of data and materials section of your manuscript, you have not provided the relevant accession numbers. Please provide the relevant accession numbers or the appropriate web links to the aforementioned data sets. Please ensure all data is already released and that any accession numbers/web links are in the ‘Availability of Data and Materials’ in their final form, and that all web links and accessions allow access to public data.

Reply: Thanks for your precious advice.
We have provided the relevant accession numbers for some of the datasets. For those with no accession numbers, we have provided the appropriate web links to the aforementioned data sets.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant
tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

Reply: Thanks for your precious advice.
We have uploaded our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.